Author’s response to reviews

Title: HBV/HIV co-infection and APOBEC3G polymorphisms in a population from Burkina Faso

Authors:

Tegwinde COMPAORE (rebecca23fr@yahoo.fr)
Birama DIARRA (diarra.birama679@gmail.com)
Maleki ASSIH (assihmalki@yahoo.fr)
Dorcas OBIRI-YEBOAH (castella.oy@gmail.com)
Serge SOUBEIGA (s.soubeiga@yahoo.fr)
Abdoul OUATTARA (ak_ouattara@yahoo.fr)
Damehan TCHELOUGOU (pascal8927@yahoo.fr)
Cyrille BISSEYE (cbisseye@gmail.com)
Didier BAKOUAN (drbakouan@yahoo.fr)
Issaka COMPAORE (pierrecomp52@yahoo.fr)
Augustine DEMBELE (dembeleaugustine@yahoo.fr)
Wendkuuni DJIGMA (florencia.djigma@gmail.com)
Jacques SIMPORE (simpore93@gmail.com)

Version: 4 Date: 07 Jun 2016

Author’s response to reviews:

Dear Editor

We are pleased to answer the reviewers’ questions about our manuscript submitted for consideration and publication in your journal. We are very grateful to the reviewers for their valuable comments and suggestions to improve our article.

Response to Reviewers:

We have taken into account all the suggestions made by the reviewer.